Author's response to reviews

Title: Physical activity patterns during pregnancy through postpartum

Authors:

Katja Borodulin (katja.borodulin@thl.fi)
Kelly R Evenson (kelly_evenson@unc.edu)
Amy H Herring (aherring@bios.unc.edu)

Version: 5 Date: 10 November 2009

Author's response to reviews:

We made one minor change; one sentence from the conclusions in the abstract has been deleted because it was not grammatically correct. We think the abstract makes sense better this way.